Author’s response to reviews

Title: A prospective study on endoscopic ultrasound for the differential diagnosis of serous cystic neoplasms and mucinous cystic neoplasms

Authors:

Lisen Zhong (Zhonglisen1005@163.com)
Ningli Chai (csxlily@163.com)
Enqiang Linghu (listen1005@163.com)
Huikai Li (13718140120@163.com)
Jing Yang (yangjing301h@163.com)
Ping Tang (tangping301h@163.com)

Version: 2 Date: 11 May 2019

Author’s response to reviews:

Dear editors and reviewers,

On behalf of my co-authors, we thank you very much for giving us an opportunity to revise our manuscript, we appreciate editor and reviewers very much for their positive and constructive comments and suggestions on our manuscript entitled “A prospective study on endoscopic ultrasound for the differential diagnosis of serous cystic neoplasms and mucinous cystic neoplasms” (BMGE-D-19-00184R1).

We have revised the manuscript, and would like to re-submit it for your consideration. We have addressed the comments raised by the reviewers. Point by point responses to the reviewers’ comments are listed below this letter.

We hope that the revised version of the manuscript is now acceptable for publication in your journal. Looking forward to hearing from you.

Thank you and best regards.

Yours Sincerely,

Enqiang Linghu
Corresponding author

Name: Enqiang Linghu

E-mail: linghuenqiang@vip.sina.com

Reply to Reviewer 3:

I recommend to add interobserver discrepancy results to the manuscript.

Answer: We have added interobserver discrepancy results to the manuscript. (Methods section, page 6, line 123-125; Results section, page 8, line 159-162)